Author's response to reviews

Title: HFE H63D mutation frequency shows an increase in Turkish women with breast cancer.

Authors:

Aysen Gunel-Ozcan (agozcan@yahoo.com)
Sibel Alyilmaz-Bekmez (alyilmazs@mynet.com)
Nilufer E Guler (nguler@hacettepe.edu.tr)
Dicle Guc (dguc@hacettepe.edu.tr)

Version: 4 Date: 21 January 2006

Author's response to reviews: see over
January 20, 2006

Re: HFE H63D mutation frequency shows an increase in Turkish women with breast cancer

Dear Editor,

Thank you very much for your letter of January 12, 2006 and the re-comments of the reviewers. Enclosed please find authors’ reply to the re-comments of the first reviewer and the manuscript altered accordingly.

We hope that the revised manuscript will meet the requirements for publication

Yours sincerely,

Aysen Gunel-Ozcan, MD, Ph.D.
Authors’ reply to the re-comments of the reviewer:

In Specific:

**To Referee 1 (Ernest Beutler)**

Major Compulsory Revisions

1. To comply with the comments of the reviewer, re: Referee 1 Major re-revision #2, necessary revisions have been made in the Discussion section, paragraph 4, paragraph line 1-6.

2. Suggested caution, re: Referee1 Major re-revision #7, have been addressed in the Discussion section, paragraph 2, paragraph line 22-26.